Author's response to reviews

Title: Gaps in Indigenous disadvantage not closing: A census cohort study of social determinants of health in Australia, Canada, and New Zealand from 1981-2006

Authors:

Francis Mitrou (francism@ichr.uwa.edu.au)
Martin Cooke (cooke@uwaterloo.ca)
David Lawrence (dlawrence@ichr.uwa.edu.au)
David Povah (d.povah@abs.gov.au)
Elena Mobilia (elena.mobilia@abs.gov.au)
Eric Guimond (Eric.Guimond@aadnc-aandc.gc.ca)
Stephen R Zubrick (steve@ichr.uwa.edu.au)

Version: 2 Date: 12 February 2014

Author's response to reviews:

Response to reviewers was provided in the covering letter